Author’s response to reviews

Title: An unexpected, mild phenotype of glucocorticoid resistance associated with glucocorticoid receptor gene mutation Case report and review of the literature

Authors:

Ágnes Molnár (agimolnar88@gmail.com)
Attila Patocs (patocs.attila@med.semmelweis-univ.hu)
Gábor Nyirò (nyirogabor1@gmail.com)
Károly Rácz (racz@bel2.sote.hu)
Miklós Tóth (totmik@gmail.com)
Beatrix Sármán (sarmanb@hotmail.com)

Version: 1 Date: 04 May 2017

Author’s response to reviews:

- Consent for publication

We appreciate that no images or videos are included in the manuscript, however, as the whole case report constitutes identifiable information on the subject investigated, consent for publication must have been obtained. Please clarify whether consent was obtained and amend this statement accordingly.

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English.

Dear Editor,

Based on your comments we included a statement about the Consent form. All the authors and the patient signed the Consent for publication.

A native English scientist reviewed our manuscript.